15-859(B) Machine Learning Theory

Homework # 4.5 Due: March 27, 2008

Some time within the next few weeks, come talk with me (in person or by email) about your plans for your class project. Typical projects might be:

• Posing and thinking about some theoretical machine learning problem, or taking two similar problems and seeing how they relate to each other.

• Reading one or two papers that go deeper into some topic discussed in or related to the course, and explaining what they are about.

• Or an experimental project related to algorithms or questions from the course. You should plan on giving a short (10-min) presentation on your project in the last week of class, and handing in a short writeup (say, 5 pages) by April 30.